Author's response to reviews

Title: External decontamination of wild leeches with hypochloric acid

Authors:

Atakan Aydin (atakanaydin@yahoo.com)
Hasan Nazik (hasannazik01@yahoo.com)
Samet V Kuvat (sametkuvat@yahoo.com)
Nezahat Gurler (gurlern@istanbul.edu.tr)
Betigul Ongen (ongenb@istanbul.edu.tr)
Serdar Tuncer (serdartuncer@hotmail.com)
Emre Hocaoglu (emrehocaoglu@hotmail.com)
Sinan N Kesim (snkesim@yahoo.com)

Version: 3 Date: 7 July 2004

Author's response to reviews:

Editor in Chief
BMC Infectious Diseases
Dear Sir,
Thank you for your e-mail indicating the acceptance of the manuscript "External decontamination of wild leeches with hypochloric acid MS: 1638464305287455" for publication in BMC Infectious Diseases. We confirm that we obtained consent from the patient to use their information and images for publication on the current paper.
We revised the document according to the formatting changes requested.
1- The word "Title" and all unnecessary capitalisation were removed.
2- Affiliations and superscript numbers were arranged accordingly.
3- We removed the author's qualifications.
4- Same font size was used in all text.
5- "None declared" was written for competing interests section.
6- We used initials for authors' contributions.
7- Journal names were cited in full and all authors were listed.
8- Tables were arranged to fit A4 pages in portrait format.
Thank you again for your cooperation and understanding.
Respectfully Yours
Atakan Aydin